Author's response to reviews

Title: A prospective randomised trial comparing nasogastric with intravenous hydration in children with bronchiolitis (protocol) The comparative rehydration in bronchiolitis study (CRIB)

Authors:

Ed Oakley (Ed.oakley@southernhealth.org.au)
Franz E Babl (franz.babl@rch.org.au)
Jason Acworth (Jason.Acworth@health.qld.gov.au)
Meredith Borland (Meredith.Borland@health.wa.gov.au)
David Krieser (David.Krieser@wh.org.au)
Jocelyn Neutze (Jocelyn.Neutze@middlemore.co.nz)
Theane Theophilos (theane.theophilos@mcri.edu.au)
Susan Donath (Susan.donath@mcri.edu.au)
Mike South (mike.south@rch.org.au)
Andrew Davidson (andrew.davidson@rch.org.au)
Paediatric Research in Emergency Departments International Colla (PREDICT) (ed.oakley@southernhealth.org.au)

Version: 5 Date: 6 May 2010

Author's response to reviews: see over
7/5/2010

Dear BMC production team

Re: The comparative rehydration in bronchiolitis study (CRIB)
Journal: BMC Pediatrics
MS ID: 1582445646340298

Formatting changes as below have been made.

Major Revisions

1. Affiliations have been written in full as requested.

2. The research group name has been included in the authors. PLEASE NOTE however that the full group title – Paediatric Research in Emergency Departments International Collaborative – did not fit in the “first name” box and the final few letters are therefore omitted.

3. Names of the members of the group have been removed from the title page and added to the acknowledgements (along with the affiliated institutions).

4. The group’s contribution has been added to the acknowledgements also.

5. Abstract headings have been amended.

6. Manuscript headings have been amended (subheadings have been retained).

7. All references are now cited in numerical order.

8. Tables 1 and 3 have been removed and added as additional files, and references in text as figures 2 and 3.

9. Table 2 has been remade into a single table (not in two parts).

10. Figure 1 is now cited in the appropriate place – sorry for this omission.

Minor Revisions

1. All tracking and comments have been removed.

2. Figure names have been removed from the figure and additional files documents.

3. Spelling and grammar have been checked.

Thank you for accepting this article for publication

Kind Regards
Ed Oakley